Author's response to reviews

Title: COMBiT: Protocol of a randomised comparison trial of COMbined modified constraint induced movement therapy and Bimanual Intensive Training with distributed model of standard upper limb rehabilitation in children with congenital hemiplegia.

Authors:

Roslyn N Boyd (r.boyd@uq.edu.au)
Jenny Ziviani (j.ziviani@uq.edu.au)
Leanne Sakzewski (leanne.sakzewski@uqconnect.edu.au)
Laura Miller (l.miller2@uq.edu.au)
Stephen Rose (Stephen.Rose@csiro.au)
Ross Cunnington (r.cunnington@uq.edu.au)
David A Abbott (d.abbott@brain.org.au)
Robert Ware (r.ware@sph.uq.edu.au)
Andrea Guzzetta (a.guzzetta@inpe.unipi.it)
Richard Macdonell (Richard.Macdonell@austin.org.au)
Graeme Jackson (g.jackson@brain.org.au)

Version: 2 Date: 30 April 2013

Author's response to reviews: see over